Author's response to reviews

Title: The Association between Oxcarbazepine-induced Maculopapular Eruption and HLA-B Alleles in a Northern Han Chinese Population

Authors:

Lv yudan (lvyudan@sina.com)

Version: 2 Date: 10 January 2013

Author's response to reviews: see over
1. Kindly include all the authors in the submission system because at present there is only one author on the system.
   
   R: I have added in submission system.

2. Competing interests - A competing interest exists when your interpretation of data or presentation of information may be influenced by your personal or financial relationship with other people or organizations. Authors must disclose any financial competing interests; they should also reveal any non-financial competing interests that may cause them embarrassment were they to become public after the publication of the manuscript.

   R: I have declared no financial or non-financial competing interests in red in the last part of Acknowledgements in manuscript.

3. Authors' contributions - In order to give appropriate credit to each author of a paper, the individual contributions of authors to the manuscript should be specified in this section.

   R: I have interpreted every author’s contribution to this manuscript in another word document uploaded in supplement materials.

4. Please be informed that table 1 contains identifiable patient information and we cannot publish this as it is. You will need to amend the table.

   R: I have made patients’ information in the Table identifiable.